# Longer Life Foundation Final Report History of Mental Health Treatment and Survival Among Newly Diagnosed Women with Breast Cancer Principal Investigator: Mark Walker, Ph.D.

# Abstract

Research has shown an association between psychological status and survival in some cancers. This study examined current distress, history of mental health problems (MHP), and history of mental health treatment (MHT) as predictors of overall survival among newly diagnosed patients with breast cancer.

**Methods:** Eligible patients (N=302) were women, 18 and older, recently diagnosed with stage 0 – IV breast cancer, and had completed psychosocial surveys between May 2000 and October 2003. The survey included self-reported history of MHP and MHT. Survey data were linked with death records from the state departments of health, and with data abstracted from the outpatient medical record. Planned methods included Cox regression models of overall survival among patients with and without MHP and MHT. Models of MHP and MHT were constructed separately. Covariates in each model included stage of disease, comorbidities, treatment adherence, health behaviors, and a composite measure of general distress.

**Results:** Patients were 73.5% Caucasian, with mean age of 53.5 years (SD=11.9), and 84.7% had stage II or earlier disease. Thirty-four patients (11.3%) had died by early 2007. Overall, 20.2% of patients reported a history of MHP, and 24.8% reported a history of MHT. Results for models of MHP and MHT showed that longer survival was associated with higher education, less comorbid disease, earlier stage of disease, and less distress at diagnosis. MHP and MHT both showed patterns in which patients with a positive history showed longer survival. The effect was nonsignificant for MHP, but significant for MHT (p=.016). Subsequent evaluation showed that the number of predictors used in the models was inappropriate for the number of observed events (deaths). Because the primary aims require covariates for control of relevant patient characteristics, revised statistical modeling cannot easily address this limitation. Results should be interpreted with caution.

**Conclusions:** Findings support a link between either mental health treatment or treatment seeking, and breast cancer survival, and are at odds with some existing research. However, findings are subject to important methodological limitations.

#### Lay Summary

Research has shown that for some types of cancer, patients who are more psychologically distressed tend not to survive as long. However, little research has been done that examines whether mental health history might be related to survival. This study examined two characteristics of mental health history as predictors of survival among newly diagnosed patients with breast cancer. The two characteristics are history of mental health problems, and history of mental health treatment. Eligible patients included 302 women, 18 and older, who were recently diagnosed with stage 0 – IV breast cancer. All of the patients had completed surveys related to emotional distress and mental health history at the time of their diagnosis. Surveys were completed between May 2000 and October 2003. The survey information was linked with death records from the state departments of health, and with other information drawn from each patient's medical record. The analysis took several other variables into

account, including stage of disease, other medical conditions, whether patients followed their doctor's treatment recommendations, and the patients level of emotional distress at the time of diagnosis.

Results showed that 73.5% of patients were Caucasian, and that patients were an average of 53.5 years of age. Most (84.7%) had stage II or earlier disease. Of the 302 patients, 34 (11.3%) had died by the time of the analysis. Approximately 1 in 4 patients reported a history of mental health problems, and 1 in 5 reported a history of mental health treatment. The analysis showed that patients tended to live longer if they had more education, fewer medical problems of other kinds, earlier stage disease, and if they were less distressed at diagnosis. All of these findings were expected. However, the results also showed that patients who reported having had mental health problems and mental health treatment were likely to live longer. These findings were not expected. The effect of mental health treatment was stronger than the effect of mental health problems, and did not appear likely to be just a chance finding. However, careful outside review of the methods concluded that the research tried to draw too many conclusions with too few patients. Because of this, the findings should be viewed as interesting but not conclusive in any way. They should not lead to any strong conclusions about the importance of mental health history. They should instead suggest possible avenues for research in the future.

#### Introduction

A large body of research has investigated psychological functioning in patients with cancer. Much of this concerns the emotional impact of cancer <sup>1-4</sup>, or the role of emotional distress during/after cancer treatment <sup>5,6</sup>. However, at least one study has examined the impact of a history of emotional problems at the time of diagnosis, and found that a history of depression was associated with shorter survival <sup>7</sup>.

There are several plausible processes by which current or past psychological problems could influence survival. First, patients with a history of mental health problems (MHP) might delay seeking medical treatment after onset of symptoms <sup>8</sup>. Second, patients with a history of MHP may be more likely to engage in unhealthy behavior, which might in turn affect survival <sup>9-12</sup>. Third, patients experiencing emotional distress or those with a history of MHP may be less likely to adhere to recommended treatment <sup>3,13</sup>, which might lead to a poorer outcome <sup>14-18</sup>. Fourth, emotional distress during and after cancer treatment might influence survival through a psychoneuroimmunological (PNI) pathway that affects response to treatment or course of disease <sup>19-22</sup>. If previous MHP is associated with distress during and after diagnosis, then history of MHP might be associated with survival.

If psychological functioning influences survival through one or more of these paths, then interventions that reduce distress or enhance psychological functioning might also improve survival. In fact, most studies that have investigated interventions to enhance psychological functioning among cancer patients have failed to show an effect on survival <sup>6,23-26</sup>. Groenvold and colleagues theorize that the association of emotional distress and cancer survival is attributable to a shared common cause—an underlying robustness that reduces psychological distress and enhances resistance to breast cancer <sup>27</sup>. This robustness model would account for the prognostic significance of emotional distress and quality of life, but also the nonsignificant effect of psychosocial interventions in previous research.

The current study sought to explore further these issues by examining current distress at diagnosis, and self-reported history of MHP and history of mental health treatment (MHT) among newly diagnosed patients with breast cancer. We hypothesized that current distress, a history of MHP, and a history of MHT would be associated with shorter survival.

### Methods

<u>Setting and Participants:</u> Participants for this study were adult women (18 years and older), who were seen in the Breast Health Center in the Alvin J. Siteman Cancer Center at Barnes-Jewish Hospital and Washington University School of Medicine between May 2000 and October 2003. Inclusion was limited to patients who underwent a routine psychosocial evaluation as part of their outpatient visit, had not previously been seen through the Breast Health Center, and had not yet had treatment other than biopsy. The sample was also limited to Missouri and Illinois residents, the primary states served by the Siteman Cancer Center.

<u>Procedure:</u> Data collection for this study involved the abstraction and collection of data from four existing data sources, and the linking of databases. The primary data source was a psychosocial survey database developed as part of routine clinical care in the Breast Health Center at Siteman Cancer Center. The database included records from self-reported survey measures of psychosocial functioning and mental health history administered during the first 1 - 4 weeks after the diagnosis of breast cancer. This database was linked by social security number with three data sources: 1) vital records obtained from the Missouri Department of Health and Senior Services and Illinois Department of Health; 2) summary psychosocial data abstracted from psychosocial evaluation reports of each patient who completed the psychosocial survey; and 3) medical and adherence data abstracted from the outpatient medical record. This study was conducted following Washington University School of Medicine's guidelines for ethical conduct of research involving human subjects. All study procedures were approved by the Institutional Review Boards at Washington University School of Medicine and Siteman Cancer Center.

<u>Study Measures:</u> The primary outcome for this study was overall survival. Vital records were obtained through the Missouri Department of Health and Senior Services and the Illinois Department of Public Health. Vital records searches were conducted separately for each state, with different possible latencies for appearance of death records in each database. Cases for which no record of death was found (the vast majority) were considered censored on the latest observed date of death in each state.

Individual demographic characteristics collected from the medical record included age and race/ethnicity (Caucasian vs. Minority). Disease and treatment related variables included date of diagnosis, stage of disease, comorbid medical conditions, and adherence to recommended adjuvant/neoadjuvant chemotherapy and radiation treatment. Stage of disease was coded as Stage 0 to Stage 4. The burden of comorbid medical conditions was graded using the Katz adaptation of the Charlson Index <sup>28,29</sup>. Treatment adherence was computed as the proportion of planned chemotherapy and radiation therapy treatments that were delivered during the first year after diagnosis.

The psychosocial survey collected from breast cancer patients as part of routine care following diagnosis at Siteman Cancer Center included the Center for Epidemiological Studies— Depression scale (CESD)<sup>30</sup>, the state section of the State-Trait Anxiety Inventory (STAI-s)<sup>31</sup>, and the Functional Assessment of Cancer Therapy-General measure (FACT-G)<sup>32</sup>. A

composite of the CESD, STAI-s, and Emotional Well Being subscale of the FACT-G was used as a measure of general distress in this study<sup>33</sup>.

Patient self-reported history of MHP was assessed with the single item: "Have you ever had problems with nerves, tension, depression, or other emotional problems?" Responses on the 5-point response scale were dichotomized with "infrequently" or "never," representing a negative history, and "often", "very frequently", or "all the time," coded as a positive history. Patient self-reported history of MHT was assessed with the yes-no item: "Have you ever consulted with a professional person about problems with nerves, tension, depression, or other emotional problems." Items not answered were coded as indicating a negative history. Several items were abstracted from the psychosocial evaluation report that was completed as a routine part of the patient's outpatient evaluation at diagnosis. These included marital status (married vs. not or unknown), educational level ( $\leq 12$  years vs. some college), number of sources of social support (0 - 1, > 1, unknown), and several health behaviors.

Resource limitations made it impossible to abstract the medical records of all patients on whom we had psychosocial survey results. To address this limitation, we randomly ordered patients in the database, and abstracted the records of as many as possible following the random order. This assured that disease stage, comorbidity, and adherence data would be missing completely at random. To address the missing predictor information statistically, we modeled these variables with use of "plugged" means and missingness indicators <sup>34</sup>. which allowed all cases to be included in the analyses.

Statistical Methods: Descriptive statistics were calculated for all study variables. T-tests and chi-square tests of independence were used to examine the association of mental health history variables with other individual characteristics. Cox proportional hazard regression was used to predict overall survival after diagnosis. Because history of MHP and history of MHT were moderately correlated, separate models were examined for each. Within each model, demographic characteristics, disease stage, health behaviors, social support, adherence, comorbidity burden, and general distress were planned as possible covariates. The significance of each covariate was assessed with the standard likelihood ratio tests between nested models, and the proportional hazard assumption was examined for each variable in the models <sup>35</sup>.

#### Results

Three hundred two patients completed psychosocial assessments as part of multidisciplinary treatment planning during the study period, had not been seen previously at the clinic, and

represent the final study sample. Abstraction of medical records was completed for a random subsample of 145 patients. Ages ranged from 26 to 90 (mean age = 53.5 years) and the majority were Caucasian (74%), married (58%), and had early stage disease (85%). Of patients for whom there was a record of treatment (132 of the random subsample of 145), 64.4% received radiation therapy, 58.3% received chemotherapy, and 41.7% received both. Vital records showed that 34 of the 302 patients

Table 1: History of Self-Reported Mental Health Problems						
(MHP) and Treatment (MHT).						
	History of MHT					
			Not	Total		
History of MHP	No	Yes	reported	N (%)		
Never to Infrequently	160	40	14	214 (70.9%)		
Often to All the time	17	30	14	61 (20.2%)		
Not reported	10	5	12	27 (8.9%)		
Total N (%)	187	75	40	302		
	(61.9%)	(24.8%)	(13.2%)			

(11.3%) had died during the follow-up period. Breast cancer was a primary or contributing cause of death for 21 patients.

Overall, 20.2% of patients reported a positive history of MHP. A slightly larger proportion, 24.8%, reported having received MHT. Surprisingly, there was only moderate correspondence between these groups, with 9.9% of the sample (30 of 302) reporting a history of problems and of treatment, and just over half (53%) reporting neither (see Table 1).

Educational level, alcohol use, tobacco use, and social support were unrelated to history of MHP. Unmarried patients were more likely than married patients to report a history of MHP,  $\chi^2$  (1, N = 302) = 5.582, p = .018, but no more likely to report a history of MHT. Caucasians were more likely than minorities to report a history of MHP,  $\chi^2$  (1, N = 302) = 7.137, p = .007 and of MHT,  $\chi^2$  (1, N = 302) = 7.510, p = .006.

Mental health history (MHP and MHT) was unrelated to treatment adherence and unrelated to the Charlson Index of Comorbidity. However, positive history of MHP was associated with earlier stage of disease at diagnosis, t (300) = 2.531, p = .012, (stage = 1.54 vs. 1.76 for patients with and without a history of MHP). Likewise, positive history of MHT was associated with earlier stage, t (300) = 2.187, p = .03, (stage = 1.59 vs. 1.76 for patients with and without a treatment history).

Cox regression analysis showed that history of MHP did not significantly predict survival (p = .229). Greater likelihood of survival in the sample was associated with higher education (p = .004), lower comorbidity index (p = .001), lower stage of disease (p = .003) and less current distress at the time of diagnosis (p = .001), each variable controlling for the others. Age, marital status, sources of social support, and health behaviors were nonsignificant for inclusion

in the model. Adherence to recommended treatment was also nonsignificant. Although history of MHP was not significant, the direction of the observed effect was unexpected, with patients positive on mental health history nominally more likely to survive (HR = .53).

Cox regression analysis showed that history of MHT was significant in predicting survival, with previous receipt of treatment associated with greater likelihood of survival (p = .016; see Figure 1). Covariates that were significant in the previous model were significant here (see Table 2). Minority race/ethnicity status also reached significance in this model, with

minority status predicting a lower likelihood of survival (p = .049). Other variables examined were nonsignificant for inclusion in the model.

We also examined whether mental health history (either MHP or MHT) interacted with current distress in predicting survival. Findings were nonsignificant. We had also theorized that an effect of history of MHP might be mediated by current distress.



Table 2. Cox Regression Model of Survival by History of MHT.					
Variables	HR	р	95% CI		
Race = minority (vs Cauc)	2.06	.049	(1.02 - 4.15)		
Educ <= 12 yrs (vs. some college)	0.34	.005	(0.15 - 0.77)		
Charlson comorbidity index	3.05	<.001	(1.9 - 4.89)		
Stage of disease	2.36	.005	(1.33 - 4.21)		
Current distress	1.78	.001	(1.28 - 2.49)		
History of MHT	3.22	.016	(1.09 - 9.09)		
Note. All effects are adjusted for other variables in the model.					

However, the effect of history of MHP was itself nonsignificant, whether adjusted or unadjusted, so a mediation model was not supported.

External Review: External review of the findings from this study identified methodological issues that limit the conclusions that should be drawn. The primary issue is the number of events that were observed relative to the number of covariates that were included in analysis. When too many covariates are included, overfitting results, and findings cannot be trusted. In a binary logistic regression, the number of events is the frequency of cases in the smaller of the two dependent variable categories. In a survival analysis, it is the number of terminal events. The investigators were aware of this methodological issue with respect to logistic regression, which was described in earlier literature<sup>36</sup>, but were not aware of its application to survival analysis methods, which were reported more recently<sup>37</sup>. Two possible approaches to addressing the limitation in the current study would involve drastically reducing the complexity of the model, and reanalyzing the data after the number of events had increased. The former approach has limitations, and if used, would preclude addressing the primary aims of the study. Given the relatively high rate of survival of breast cancer, the second approach is not practical.

# Discussion

This study examined the association of mental health and behavioral issues with survival among patients with breast cancer. Overall survival in the sample was associated with being Caucasian, more highly educated, having less comorbid illness, less advanced stage of disease, and being less distressed at diagnosis. Survival was also associated with having received MHT, the opposite of the effect expected. It was not significantly related to self-reported history of MHP.

This effect of a positive history of MHT suggests that, whereas psychological distress may convey risk of reduced survival, treatment of mental health problems may have a beneficial effect. However, patients were self-selected for MHT—they were not assigned. It is possible, therefore, that differences exist between patients who seek MHT and those who do not, and that it is these differences that influence survival rather than treatment itself. This interpretation of the findings would be consistent with Groenvold et al.'s robustness hypothesis<sup>27</sup>.

There were important limitations to this study, including that it was observational, that it did not examine the role of distress or of mental health care after diagnosis and treatment, and the important methodological limitation described under External Review, above. The observed effect of MHT on survival may in fact reflect statistical overfitting, but indicators of such overfitting are not apparent. Specifically, the effect of mental health history was a primary focus of the study rather than one 'discovered' during data mining. Also, the observed effect was present in univariate analysis, and was essentially unchanged when controlling for other significant covariates, including stage of disease, comorbidities, and current distress. Given these considerations, the finding that mental health treatment, or the propensity to seek treatment, may convey benefit, should be considered worthy of further study.

# References

- 1. U.S. Department of Health and Human Services. Depression and Cancer. Bethesda, MD: National Institutes of Health; 2002.
- 2. Stark D, Kiely M, Smith A, et al. Anxiety disorders in cancer patients: their nature, associations, and relation to quality of life. *J Clin Oncol.*2002;20(14):3137-3148.
- 3. McDaniel JS, Musselman DL, Nemeroff CB. Cancer and depression: Theory and treatment. *Psychiatric Annals.* May 1997;27(5):360-364.
- 4. Moyer A, Salovey P. Psychosocial sequelae of breast cancer and its treatment. *Ann Behav Med.*1996;18(2):110-125.
- 5. Spiegel D, Bloom JR, Kraemer HC, et al. Effect of psychosocial treatment on survival of patients with metastatic breast cancer. *Lancet*.1989;2(8668):888-891.
- 6. Goodwin PJ, Leszcz M, Ennis M, et al. The effect of group psychosocial support on survival in metastatic breast cancer. *N Engl J Med.* Dec 13 2001;345(24):1719-1726.
- 7. Stommel M, Given BA, Given CW. Depression and functional status as predictors of death among cancer patients. *Cancer*.2002;94(10):2719-2727.
- 8. Desai MM, Bruce ML, Kasl SV. The effects of major depression and phobia on stage at diagnosis of breast cancer. *Int J Psychiatry Med*.1999;29(1):29-45.
- 9. Stunkard AJ, Faith MS, Allison KC. Depression and obesity. *Biol Psychiatry.* Aug 2003;54(3):330-337.
- 10. Johnson EO, Rhee SH, Chase GA, et al. Comorbidity of depression with levels of smoking: An exploration of the shared familial risk hypothesis. *Nicotine Tob Res.* Dec 2004;6(6):1029-1038.
- 11. Fergusson DM, Goodwin RD, Horwood LJ. Major depression and cigarette smoking: Results of a 21-year longitudinal study. *Psychol Med.* Nov 2003;33(8):1357-1367.
- 12. Dierker LC, Avenevoli S, Stolar M, et al. Smoking and depression: An examination of mechanisms of comorbidity. *Am J Psychiatry.* Jun 2002;159(6):947-953.
- 13. Goldberg RJ. Systematic understanding of cancer patients who refuse treatment. *Psychotherapy & Psychosomatics.* Jul 1983;39(3):180 -189.
- 14. Wright S. Patient satisfaction in the context of cancer care. *Irish Journal of Psychology*.1998;19(2-3):274-282.
- 15. Cathcart CS, Dunican A, Halpern JN. Patterns of delivery of radiation therapy in an innercity population of head and neck cancer patients: an analysis of compliance and end results. *Journal of Medicine*.1997;28(5-6):275-284.
- 16. Li BD, Brown WA, Ampil FL, et al. Patient compliance is critical for equivalent clinical outcomes for breast cancer treated by breast-conservation therapy. *Ann Surg*.2000;231(6):883-889.
- 17. Hortobagyi GN, Ames FC, Buzdar AU, et al. Management of stage III primary breast cancer with primary chemotherapy, surgery, and radiation therapy. *Cancer*.1988;62(12):2507-2516.
- 18. Stadler P, Putnik K, Kreimeyer T, et al. Split course hyperfractionated accelerated radiochemotherapy (SCHARC) for patients with advanced head and neck cancer: Influence of protocol deviations and hemoglobin on overall survival, a retrospective analysis. *BMC Cancer*.2006;6:279.
- 19. Spiegel D, Giese-Davis J. Depression and cancer: mechanisms and disease progression. *Biol Psychiatry.* Aug 1 2003;54(3):269-282.
- 20. Antoni MH. Psychoneuroendocrinology and psychoneuroimmunology of cancer: Plausible mechanisms worth pursuing? *Brain Behav Immun.* Feb 2003;17 Suppl 1:S84-91.
- 21. Kiecolt-Glaser JK, Robles TF, Heffner KL, et al. Psycho-oncology and cancer: psychoneuroimmunology and cancer. *Ann Oncol.*2002;13 Suppl 4:165-169.

- 22. Reiche EM, Nunes SO, Morimoto HK. Stress, depression, the immune system, and cancer. *Lancet Oncol.* Oct 2004;5(10):617-625.
- 23. Cunningham AJ, Edmonds CV, Jenkins GP, et al. A randomized controlled trial of the effects of group psychological therapy on survival in women with metastatic breast cancer. *Psychooncology.* Nov-Dec 1998;7(6):508-517.
- 24. Edelman S, Lemon J, Bell DR, et al. Effects of group CBT on the survival time of patients with metastatic breast cancer. *Psychooncology.* Nov-Dec 1999;8(6):474-481.
- 25. Kissane DW, Love A, Hatton A, et al. Effect of cognitive-existential group therapy on survival in early-stage breast cancer. *J Clin Oncol.* Nov 1 2004;22(21):4255-4260.
- 26. Goodwin PJ. Support groups in advanced breast cancer. *Cancer.* Dec 1 2005;104(11) Suppl:2596-2601.
- 27. Groenvold M, Petersen MA, Idler E, et al. Psychological distress and fatigue predicted recurrence and survival in primary breast cancer patients. *Breast Cancer Res Treat.* Oct 2007;105(2):209-219.
- 28. Charlson ME, Pompei P, Ales KL, et al. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. *J Chronic Dis.*1987;40(5):373-383.
- 29. Katz JN, Chang LC, Sangha O, et al. Can comorbidity be measured by questionnaire rather than medical record review? *Med Care.* Jan 1996;34(1):73-84.
- 30. Radloff L. The CES-D Scale: A self-report depression scale for research in the general population. *Applied Psychological Measurement*.1977;1(3):385-401.
- 31. Spielberger CD, Gorsuch RL, Lushene RE. STAI manual for the State-Trait Anxiety Inventory. Palo Alto: Consulting Psychologists Press; 1970.
- 32. Cella DF, Tulsky DS, Gray G, et al. The Functional Assessment of Cancer Therapy (FACT) scale: Development and validation of the general measure. *J Clin Oncol*.1993;11(3):570-579.
- 33. Deshields T, Reschke A, Walker MS, et al. Psychological status at time of diagnosis and patient's ratings of cosmesis following radiation therapy for breast cancer. *Journal of Psychosocial Oncology*.2007;25:103-116.
- 34. Cohen J, Cohen P, West SG, et al. *Applied multiple regression / correlation analysis for the behavioral sciences*. 3rd ed. Hillsdale, NJ: Lawrence Erlbaum Assoc.; 2002.
- 35. Hosmer DW, Lemeshow S. *Applied Survival Analysis: Regression Modeling of Time to Event Data*. New York, NY: John Wiley & Sons, Inc.; 1998.
- 36. Peduzzi P, Concato J, Kemper E, et al. A simulation study of the number of events per variable in logistic regression analysis. *Journal of clinical epidemiology.* Dec 1996;49(12):1373-1379.
- 37. Babyak MA. What you see may not be what you get: a brief, nontechnical introduction to overfitting in regression-type models. *Psychosom Med.* May-Jun 2004;66(3):411-421.